Author's response to reviews

Title: Anesthetic management in a patient with giant growing teratoma syndrome: a case report

Authors:

Nobuko Ohashi (ww81854@sa2.so-net.ne.jp)
Hidekazu Imai (hdkzimi@med.niigata-u.ac.jp)
Toshiyuki Tobita (zz-top@med.niigata-u.ac.jp)
Hideaki Ishii (hishii@med.niigata-u.ac.jp)
Hiroshi Baba (baba@med.niigata-u.ac.jp)

Version: 2 Date: 24 August 2013

Author's response to reviews:

August 24th, 2013

Dear Editor-in-Chief, Journal of Medical Case Reports, Michael Kidd

We are submitting a manuscript entitled “Anesthetic management in a patient with giant growing teratoma syndrome” for consideration for publication in the Journal of Medical Case Reports.

We confirm that the work in the manuscript is original and that the manuscript has not been published or submitted for publication elsewhere. And this is the first report with anesthetic management in a patient with growing teratoma syndrome with a solid mass type. Because anesthetic management during surgical resection of a giant ovarian tumor of the cystic type has been reported, but there are no similar reports for resection of solid mass type giant ovarian tumors. It will advance our understanding of a particular disease in anesthetic management and have a broader clinical impact across more than one area like anesthesia, surgery and intensive care unit.

All the authors have read the manuscript and all have approved submission of the manuscript for publication. None of the authors have a conflict of interest regarding the work in the manuscript.

Thank you for your consideration of the manuscript. We look forward to your reply.

Sincerely,

Corresponding Author: Nobuko Ohashi

Co-Authors: Hidekazu Imai, Toshiyuki Tobita, Hideaki Ishii, Hiroshi Baba